Annals of Internal Medicine

ARTICLE

Changes in Triglyceride Levels and Risk for Coronary Heart Disease in Young Men

Amir Tirosh, MD, PhD; Assaf Rudich, MD, PhD; Tzippora Shochat, MSc; Dorit Tekes-Manova, MD; Eran Israeli, MD; Yaakov Henkin, MD; Ilan Kochba, MD; and Iris Shai, RD, PhD

Background: Current triglyceride levels might be only a weak predictor of risk for coronary heart disease (CHD).

Objective: To assess the association between changes over time in fasting triglyceride levels and CHD risk in young adults.

Design: Follow-up study over 5.5 years after 2 measurements of fasting triglycerides 5 years apart.

Setting: The Staff Periodic Examination Center of the Israel Defense Forces, Zrifin, Israel.

Patients: 13 953 apparently healthy, untreated, young men (age 26 to 45 years) with triglyceride levels less than 3.39 mmol/L (<300 mg/dL).

Measurements: Two triglyceride measurements (at enrollment [time 1] and 5 years later [time 2]), lifestyle variables, and incident cases of angiography-proven CHD.

Results: Within 5.5 years, 158 new cases of CHD were identified. The multivariate model was adjusted for age; family history; fasting glucose; high-density lipoprotein cholesterol; blood pressure; body mass index; and changes between time 1 and time 2 in body mass index, physical activity, smoking status, and habit of eating breakfast. Investigators categorized triglyceride levels according to low,

recent meta-analysis (1) and most published papers suggest a moderate association between fasting triglyceride levels and coronary heart disease (CHD) (2-11). Of the lipid fractions, the triglyceride-rich very-low-density lipoprotein particle is probably the most sensitive to lifestyle modification (8). For example, estimates from meta-analyses suggest that for every 4.5 kg (approximately 10 lb) of stable weight reduction, triglyceride levels decrease by at least 0.068 mmol/L (≥ 6 mg/dL) (12). Accordingly, a considerable increase in the proportion of hypertriglyceridemic patients accompanies the obesity epidemic (6, 9). In addition, aerobic exercise, independent of weight loss, has been shown to modestly reduce triglyceride levels in a dosedependent fashion (13). Hence, when assessing the risk associated with triglyceride levels, triglyceride measurement at a single time point (typically at enrollment) may not be a reliable indicator of a person's long-term triglyceridemia during follow-up. Whether changes in triglyceride levels over time can improve cardiovascular risk assessment is largely unknown, particularly in young adults, in whom information on the association between triglycerides and CHD is not available (1).

For 13 953 apparently healthy young adult men (mean age, 32 years; range, 26 to 45 years) from the MELANY (Metabolic, Lifestyle, and Nutrition Assessment in Young Adults) study (14), we obtained 2 measurements intermediate, and high tertiles (as measured at time 1 and time 2 [expressed as tertile at time 1/tertile at time 2]). The risk for CHD in men with high-tertile triglyceride levels at time 1 changed depending on the tertile at time 2 (hazard ratios, 8.23 [95% CI, 2.50 to 27.13] for high/high, 6.84 [CI, 1.95 to 23.98] for high/intermediate, and 4.90 [CI, 1.01 to 24.55] for high/low, compared with the stable low/low group). The risk for CHD in men with low-tertile levels at time 1 also changed depending on the tertile at time 2 (hazard ratios, 3.81 [CI, 0.96 to 15.31] for low/intermediate and 6.76 [CI, 1.34 to 33.92] for low/high, compared with the stable low/low group).

Limitations: Participants were healthy and had a low incidence rate of CHD. The study was observational.

Conclusions: Two triglyceride measurements obtained 5 years apart may assist in assessing CHD risk in young men. A decrease in initially elevated triglyceride levels is associated with a decrease in CHD risk compared with stable high triglyceride levels. However, this risk remains higher than in those with persistently low triglyceride levels.

Ann Intern Med. 2007;147:377-385. For author affiliations, see end of text. www.annals.org

of fasting serum triglycerides and lifestyle variables 5 years apart and followed for incident cases of angiographyproven CHD. Here, we estimate the effect of baseline triglyceride levels (time 1) and changes (between time 1 and time 2) in triglyceride levels on CHD risk.

METHODS

The MELANY Study

The MELANY study, which was designed to investigate risk factors for common diseases in young adults, is being conducted at the Israel Defense Forces Staff Periodic Examination Center (SPEC), Zrifin, Israel. All career service personnel are evaluated every 5 years between 25 and

See also:

Print

Editors' Notes	 378
Editorial comment.	 425
Summary for Patients	 I-45

Web-Only

Conversion of graphics into slides Audio summary

ARTICLE | Changes in Triglyceride Levels and Coronary Risk in Young Men

Context

Whether the serum triglyceride level is an independent risk factor for coronary heart disease (CHD) is not clear.

Contribution

The authors measured triglyceride levels and performed stress electrocardiographies 5 years apart on 13 593 young Israeli male career soldiers and did coronary angiography if the stress test was abnormal. Triglycerides and change in triglyceride levels strongly predicted incident CHD after adjustment for known CHD risk factors and lifestyle. Decreases in triglyceride levels were associated with adoption of a healthier lifestyle and lower CHD risk. The lowest CHD risk occurred when triglyceride levels remained low.

Caution

The participants were healthy male soldiers.

Implication

In healthy young men, triglycerides and changes in triglyceride levels are an independent CHD risk factor.

—The Editors

35 years of age and every 3 years thereafter until they are discharged from service. A computerized database established in 1992 is the source of data for the MELANY study. At each SPEC visit, participants complete a detailed questionnaire that assesses demographic, nutritional, lifestyle, and medical factors. Thereafter, blood samples are drawn after a 14-hour fast and analyzed. A trained medical technician measures height, weight, and blood pressure (by mercury sphygmomanometers), and a physician at the center performs a complete physical examination.

Inclusion and Exclusion Criteria

We included apparently healthy men 26 to 45 years of age who had fasting triglyceride levels less than 3.39 mmol/L (<300 mg/dL) at their initial SPEC visit. We used the cutoff value of 3.39 mmol/L (300 mg/dL) because SPEC persons with greater triglyceride values are subjected to additional interventions-nutritional, pharmacologic, or both. Of 15 155 men age 26 to 45 years, 1202 were excluded because they had diabetes (type 1 or 2) (n = 227) or CHD (n = 17) at baseline; had fasting triglyceride levels of 3.39 mmol/L or greater (\geq 300 mg/dL) (n = 676); or were receiving long-term medications (n = 282), including lipid-lowering medications. Therefore, for analysis of the association between baseline triglyceride levels and incident CHD, we included 13 953 men. In studying the effect of changes in triglyceride levels on CHD, we excluded an additional 413 men who did not have a triglyceride measurement from the second SPEC visit (n = 363) or had received a diagnosis of diabetes (n = 38) or CHD (n =12) between the first and second SPEC visit or during the

second visit. Women were not included because the number of new cases of CHD in women was too small to facilitate meaningful analysis.

Outcome Definition

The outcome definition of the study was clinically significant CHD (angiography-proven stenosis >50% in at least 1 coronary artery) or fatal or nonfatal myocardial infarction (MI). At each sequential SPEC visit, all Israel Defense Forces military personnel older than 35 years of age who were participating in the current analysis had a treadmill exercise test (Bruce protocol [15]) in the presence of a board-certified cardiologist. End points for the exercise test were clinically significant ST-segment depression (>2 mm in 2 contiguous leads, measured 80 ms after the J point), intolerable symptoms of angina and exhaustion, or achievement of the target heart rate. All cases with a pathologic stress test were referred for coronary angiography. In participants with a borderline stress test, or when participants reported angina symptoms without diagnostic electrocardiographic changes, stress perfusion imaging with thallium-201 was performed. Those with a pathologic thallium-201 cardiac scan underwent coronary angiography. All Israel Defense Forces personnel obtain primary care between scheduled SPEC visits at designated military clinics, and all medical records are stored in the same central database, thereby facilitating ongoing, tight, and uniform follow-up. Individuals presenting with symptoms of angina, MI, or both between SPEC visits were also referred for coronary angiography after consultation with a boardcertified cardiologist.

Laboratory Methods

Investigators performed biochemical analyses of fresh blood samples in an adjacent core laboratory facility that handles 1.2 million samples per year. The laboratory is authorized to perform tests according to the International Organization for Standardization standard 9002. The United Kingdom National External Quality Assessment Service, Sheffield, United Kingdom, performed periodic assessment of quality control on a regular basis. All lipid levels were directly measured, except for low-density lipoprotein (LDL) levels, which were calculated. Investigators measured all biochemical markers by using a BM/ Hitachi 917 automated analyzer (Boehringer, Mannheim, Germany).

Statistical Analysis

For the primary analysis, we included 13 953 untreated, apparently healthy young men with triglyceride levels lower than 3.39 mmol/L (<300 mg/dL). We used a general linear model to assess the age-adjusted means and proportions of the population's characteristics across quintiles of triglycerides and to fit the median of the quintiles as a continuous variable to estimate the trend of variables across quintiles (Table 1). We conducted Cox proportional hazards analysis during the 10.5-year follow-up to estimate the hazard ratios and 95% CIs for the development of

Characteristic	Quintile 1 (<i>n</i> = 2844)	Quintile 2 (<i>n</i> = 2739)	Quintile 3 (<i>n</i> = 2789)	Quintile 4 (<i>n</i> = 2795)	Quintile 5 (<i>n</i> = 2786)	P Value for Trend			
Triglyceride level						-			
Mean (SD)									
mmol/L	0.59 (0.11)	0.88 (0.08)	1.17 (0.09)	1.57 (0.14)	2.39 (0.40)				
mg/dL	52 (9.6)	78 (6.9)	104 (8.4)	139 (12.6)	212 (35.7)				
Median (25th, 75th percentiles)									
mmol/L	0.60 (0.51, 0.68)	0.88 (0.81, 0.95)	1.17 (1.10, 1.25)	1.56 (1.45, 1.69)	2.30 (2.05, 2.66)				
mg/dL	53 (45, 60)	78 (72, 84)	104 (97, 111)	138 (128, 150)	204 (182, 236)				
Range									
mmol/L	0.34–0.75	0.76–1.02	1.03–1.34	1.35–1.84	1.85–3.38				
mg/dL	30–66	67–90	91–119	120–163	164–299				
Mean age (SD), y	31.0 (4.8)	31.7 (5.1)	32.3 (5.3)	33.0 (5.4)	33.9 (5.2)	< 0.001			
	10.1	10.0	10.1	10.6	40.7	0.50			
Mean follow-up, y	10.4	10.3	10.1	10.6	10.7	0.58			
Eamily history of CHD %+	16.1	15.2	167	15.0	10.0	0.22			
Failing flistory of CHD, $\sqrt{61}$	24 (2 1)	24 7 (2 4)	25.5 (2.5)	10.9 26.2 (2.5)	10.0	< 0.001			
Mean blood prossure (SD) mm Hg	24 (5.1)	24.7 (5.4)	29.9 (5.9)	26.5 (5.5)	27.3 (3.0)	<0.001			
Systolic	117 (11 4)	119 (11 6)	120 (12 2)	121 (12 3)	122 (12 6)	<0.001			
Diastolic	75 (8 4)	763 (86)	77 1 (9 1)	77.8 (9.0)	78 9 (9 7)	< 0.001			
Arterial pressure	88.4 (8.6)	89.2 (8.7)	90.5 (9.4)	91 2 (9 3)	92 3 (9 8)	< 0.001			
Smoking status %	00.4 (0.0)	09.2 (0.7)	50.5 (5.4)	51.2 (5.5)	52.5 (5.0)	<0.001			
Current	22	27.7	29 7	32.9	37	< 0.001			
Former	20	18.5	22	19.8	20.5	0.81			
Physical activity, % (min/wk)‡	38 (91)	35.4 (89)	32.2 (88)	29.9 (81)	24.6 (83)	< 0.001			
,						(0.057)			
Habit of eating breakfast, %	22	19.3	17.7	17.9	16.4	0.002			
Age-adjusted biomarkers									
Mean HDL cholesterol level (SD)						< 0.001			
mmol/L	1.40 (0.29)	1.25 (0.27)	1.22 (0.25)	1.15 (0.25)	1.06 (0.25)				
mg/dL	54 (11.2)	48.2 (10.5)	47.1 (9.8)	44.3 (9.8)	40.8 (9.6)				
Mean LDL cholesterol level (SD)						< 0.001			
mmol/L	3.08 (0.72)	3.39 (0.88)	3.65 (0.89)	3.83 (0.93)	3.78 (0.95)				
mg/dL	119 (27.7)	131 (33.9)	141 (34.5)	148 (36.1)	146 (36.8)				
TC-HDL ratio (SD)	3.7 (0.9)	4.28 (1.1)	4.8 (1.3)	5.34 (1.5)	6.03 (1.6)	< 0.001			
Mean fasting plasma glucose level (SD)						0.005			
mmol/L	4.94 (0.54)	5.02 (0.53)	5.08 (0.54)	5.11 (0.56)	5.18 (0.57)				
mg/dL	89 (9.7)	90.3 (9.6)	91.4 (9.8)	92 (10.1)	93.3 (10.3)				

Table 1. Baseline (Time 1) Characteristics, by Quintile of Triglyceride Level

* "Time 1" is the first of 2 measurements of triglycerides obtained 5 years apart. BMI = body mass index; CHD = coronary heart disease; HDL = high-density lipoprotein; LDL = low-density lipoprotein; TC = total cholesterol.

+ A family history of CHD includes the presence of heart disease in a first-degree relative before age 65 y (women) or 55 y (men).

‡ Defined as engagement in physical activity for a minimum of 20 minutes, 3 times per week.

CHD (the dependent variable) according to triglyceride levels at time 1 (first measurement). In stepwise models (**Table 2**), we added the values for body mass index (BMI), high-density lipoprotein (HDL) cholesterol, and family history of CHD separately to the age-adjusted model to evaluate their potential role as confounders. In the final multivariate model, we controlled for age, BMI, HDL cholesterol, family history of CHD, fasting plasma glucose, mean arterial blood pressure, physical activity, and smoking status. Because the total cholesterol–HDL cholesterol ratio is a predictor of CHD (8), we performed a secondary analysis that included this ratio instead of HDL cholesterol in the multivariate model.

To assess the risk associated with *changes* in triglyceride levels, we analyzed data from 13 540 men who had 2 triglyceride measurements (obtained at time 1 and time 2, 5 years apart) that were available before the end of follow-up or before being censored after a diagnosis of CHD or diabetes. In the model, time 2 was considered the baseline of 5.2 years of subsequent follow-up, whereas time 1 was considered prebaseline. We cross-classified triglyceride levels at each time point by tertiles: median levels of 0.68, 1.18, and 2.08 mmol/L (60, 104, and 184 mg/dL) at time 1 and 0.79, 1.33, and 2.49 mmol/L (70, 118, and 220 mg/dL) at time 2. In parallel, we determined changes in BMI, smoking status, physical activity, and habit of eating breakfast between time 2 and time 1 (Figure 1). Next, we evaluated the joint risk attributed to triglyceride levels at time 1 and time 2, categorized according to low (bottom), intermediate, and high (top) tertiles, and we used men with triglyceride levels in the low tertile at both time 1 and time 2 as a reference group (low/low group; hazard ratio,

Variable	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	P Value for Trend
Range of triglyceride level						-
mmol/L	0.34-0.75	0.76-1.02	1.03-1.34	1.35–1.84	1.85-3.38	
mg/dL	30–66	67–90	91–119	120–163	164–299	
Follow-up, person-years	29 578	28 212	28 169	29 627	29 810	
Incident cases of CHD, n	8	13	37	42	70	< 0.001
Adjusted risk ratio (95% CI)						
Age	1.00 (reference)	1.43 (0.70–2.94)	4.48 (1.97–8.85)	5.10 (2.19–10.63)	7.06 (3.72–14.81)	< 0.001
Age and BMI	1.00 (reference)	1.26 (0.63–2.78)	4.17 (1.90–8.11)	4.03 (2.02–9.41)	6.22 (3.19–12.68)	< 0.001
Age, BMI, and HDL cholesterol level	1.00 (reference)	1.12 (0.59–2.34)	3.70 (1.81–7.57)	3.84 (1.86–8.29)	5.15 (2.84–10.02)	< 0.001
Age, BMI, HDL cholesterol level, and family history of CHD	1.00 (reference)	1.13 (0.62–2.41)	3.78 (1.96–7.60)	3.96 (1.91–8.35)	5.29 (2.93–10.17)	< 0.001
Multivariate†	1.00 (reference)	1.04 (0.56–2.30)	2.92 (1.65–6.39)	3.18 (1.72–7.24)	4.05 (2.68–8.61)	<0.001

Table 2. Hazard Ratios for Coronary Heart Disease, by Quintile of Time-1 Triglyceride Level*

* BMI = body mass index; CHD = coronary heart disease; HDL = high-density lipoprotein. † The multivariate Cox regression time-dependent model was adjusted for age, BMI, HDL cholesterol level, fasting plasma glucose level, mean arterial blood pressure (continuous variables), family history of CHD (positive, negative, or missing information), physical activity (yes, no, or missing information), and smoking status (current, noncurrent, or missing information).

1.0). To evaluate the direct association of changes in triglyceride levels (Figure 2), we used a multivariate model to further adjust for the interval between the 2 measurements and for the changes between time 2 and time 1 in BMI and lifestyle variables (physical activity, smoking, and habit of eating breakfast). We included these variables by calculating differences (Δ) in BMI and creating 3 groups of each categorical variable (smoking, physical activity, and habit of eating breakfast) based on their status at time 1 and time 2 (yes/yes, yes/no, no/yes, or no/no).

For the analysis of the association between cardiovascular risk and baseline triglyceride measurement (Table 2), follow-up consists of the interval between time 1 and the end of follow-up (or diagnosis)-that is, 10.5 years-and includes the cases that occurred between time 1 and time 2 (n = 12). For the analysis of the association between cardiovascular risk and different directionality of changes in triglycerides (Figure 2), follow-up was from the second triglyceride determination (time 2) to the end of follow-up (or diagnosis)-that is, 5.2 years-effectively converting the value at time 1 into a prebaseline variable. We used SAS statistical software, version 9.1 (SAS Institute, Cary, North Carolina), for all statistical analyses.

Role of the Funding Source

The Ben-Gurion University of the Negev, Beer-Sheva, Israel, and funds from the Israel Defense Forces National Budget provided support for the study. The funding sources had no role in the conduct of the study or in the decision to submit the manuscript for publication. The Israel Defense Forces had a role in the design of the cohort and in performing the follow-up and outcome measurements. Dr. Tirosh and Ms. Shochat had full access to all of the data in the study and take responsibility for the integrity of the data and the accuracy of the data analysis.

RESULTS

We analyzed the data of 13 953 untreated, apparently healthy men (26 to 45 years of age) with fasting serum triglyceride levels lower than 3.39 mmol/L (<300 mg/dL). At time 1, age-adjusted LDL cholesterol and fasting plasma glucose levels, BMI, blood pressure, and proportion of current smokers were more likely to increase across quintiles of triglycerides, whereas HDL cholesterol levels and the proportion of men who were physically active and who reported eating breakfast regularly were more likely to decrease (Table 1).

A total of 170 incident cases of angiography-proven CHD occurred during 145 396 person-years of follow-up (mean follow-up, 10.5 years): 163 cases were diagnosed by coronary angiography, and 7 men had nonfatal MI as their presenting symptom before their scheduled SPEC visit. No cases of fatal MI were documented during follow-up. Ageadjusted hazard ratios for CHD increased across quintiles of triglyceride level, reaching 7.06 (95% CI, 3.72 to 14.81) for the top quintile (median, 2.30 mmol/L [204 mg/dL]) versus the bottom quintile (median, 0.60 mmol/L [53 mg/ dL]) (P < 0.001 for trend) (Table 2). Further adjustment for BMI attenuated the risk value to 6.22 (CI, 3.19 to 12.68) for the top quintile. Predictably, this association was additionally attenuated after adjustment for HDL cholesterol (hazard ratio, 5.15 [CI, 2.84 to 10.02] for the top quintile) but did not change after further adjustment for family history of CHD. In a multivariate model adjusted for age, BMI, HDL cholesterol level, family history of CHD, fasting plasma glucose level, mean arterial blood pressure, physical activity, and smoking status, the association between triglyceride level and CHD remained independent from the third quintile (median, 1.17 mmol/L [104 mg/dL]), reaching a hazard ratio of 4.05 (CI, 2.68 to 8.61) for the top quintile versus the bottom quintile (P <0.001 for trend). This association remained unchanged in

a secondary analysis that included the total cholesterol– HDL cholesterol ratio, instead of HDL cholesterol level, in the multivariate model. To exclude the possibility that elevated triglyceride levels already reflected clinically unrecognized CHD at baseline, we confirmed these clinically significant findings in a secondary analysis that excluded 5 persons with CHD that was diagnosed within the first 2 years of follow-up (data not shown).

We next analyzed data from 13 540 men for whom we had triglyceride measurements at 2 time points (time 1 and time 2) 5 years apart before being censored because of a diagnosis of CHD. Within this group (Figure 1), we characterized the men whose triglyceride levels were in the low (bottom) or high (top) tertile at time 1 and either remained stable in time 2 (that is, low/low [n = 2846] or high/high [n = 2840], respectively) or had a marked change in their time-2 measurement (that is, low/high [n = 388] or high/low [n = 402], respectively). These changes in triglycerides were associated with significant alterations in BMI, physical activity, and habit of eating breakfast between the 2 time points (P < 0.05). An increase in triglyceride levels

between time 1 and time 2 (that is, low/high) was associated with increased BMI, decreased levels of physical activity, and a decrease in the proportion of men who ate breakfast. Conversely, young men in the high/low group (that is, whose triglyceride levels decreased without antidyslipidemic therapy) had a decrease in mean BMI and increase in physical activity and in the proportion of eating breakfast, consistent with the sensitivity of triglycerides to lifestyle factors. Retaining low triglyceride levels (that is, low/ low) was associated with stable physical activity levels and habit of eating breakfast and the lowest BMI and proportion of current smoking.

Consistent with the findings presented in Table 2, a multivariate model to show the relationship of triglyceride values defined in tertiles at enrollment (time 1) revealed that the risk for incident CHD was 3.59-fold (CI, 1.89 to 7.69) and 4.66-fold (CI, 2.50 to 9.86) higher (P < 0.001 for trend) in the intermediate and high tertiles of triglyceride levels, respectively, compared with the low tertile.

We further evaluated the risk for CHD according to changes of triglyceride levels over time (Figure 2 and Table 3)

Figure 1. Changes in selected lifestyle variables associated with 5-year changes, without medications, in triglyceride levels between time 1 and time 2.



Tertiles were defined as low, the bottom tertile of triglyceride level (median at time 1, 0.68 mmol/L [60 mg/dL]; median at time 2, 0.79 mmol/L [70 mg/dL]), or high, the top tertile of triglyceride level (median at time 1, 2.08 mmol/L [184 mg/dL]; median at time 2, 2.48 mmol/L [220 mg/dL]). *Engagement in physical activity for a minimum of 20 minutes, 3 times per week.

ARTICLE | Changes in Triglyceride Levels and Coronary Risk in Young Men

Figure 2. Multivariate model showing the association between fasting serum triglyceride levels obtained at 2 measurements 5 years apart and incidence of coronary heart disease.



The model was adjusted for age, family history of coronary heart disease (positive, negative, or missing information), interval between time 1 and time 2, time-1 levels of fasting plasma high-density lipoprotein cholesterol, glucose, mean arterial blood pressure, and body mass index (as continuous variables), and for changes between time 1 and time 2 in BMI, physical activity (nonactive/nonactive, nonactive/active, active/nonactive, active/active), smoking status (current/current, current/noncurrent, noncurrent/current, noncurrent/noncurrent), and habit of eating breakfast (no/no, no/yes, yes/no, yes/yes). To convert triglyceride values to mmol/L, multiply by 0.0113. Incidence rates and hazard ratios (with 95% CIs) are given in Table 3.

by using a multivariate model that was further adjusted for the changes in BMI and lifestyle variables (physical activity, smoking, and habit of eating breakfast). Young men who had low triglyceride levels at time 1 had a 3.81-fold (CI, 0.96 to 15.31) or 6.76-fold (CI, 1.34 to 33.92) greater risk for CHD within the 5.5-year follow-up if their second triglyceride measurement (at time 2) increased to intermediate or high levels, respectively, compared with the low/low group. The risk for CHD for men with low levels of triglycerides at time 2 was modified by their time-1 values, being 3.88-fold (CI, 1.04 to 22.50) and 4.90-fold (CI, 1.01 to 24.55) higher if previous triglyceride (time 1) levels were intermediate or high, respectively, compared with the low/low group. The risk for CHD in men with initially high triglyceride levels (at time 1) changed if triglyceride levels subsequently decreased to intermediate or low levels at time 2 (6.84-fold [CI, 1.95 to 23.98] or 4.90-fold [CI, 1.01 to 24.55], respectively) or remained high (8.23-fold [CI, 2.50 to 27.13]), compared with the low/ low group. These hazard ratios were independent of the *changes* in BMI and the other lifestyle variables (smoking status, physical activity, and habit of eating breakfast).

Table 3. Incidence Rates and Hazard Ratios of Coronary Heart Disease, by Tertile of Triglyceride Level and Time of Measurement*

Variable	Low: Time 1			Intermediate: Time 1				High: Time 1				
	Population at Risk, <i>n</i>	Events, n	Incidence Rate per 10 000 Persons	HR (95% CI)	Population at Risk, <i>n</i>	Events, n	Incidence Rate per 10 000 Persons	HR (95% CI)	Population at Risk, <i>n</i>	Events, n	Incidence Rate per 10 000 Persons	HR (95% CI)
Low: Time 2	2846	4	14.1	1	1230	8	65	3.88 (1.04–22.50)	402	5	124.4	4.90 (1.01–24.55)
Intermediate: Time 2	1266	7	55.3	3.81 (0.96–15.31)	1992	23	115.5	5.40 (1.57–18.63)	1271	20	157.4	6.84 (1.95–23.98)
High: Time 2	388	6	154.6	6.76 (1.34–33.92)	1305	29	222.2	9.06 (2.65–30.90)	2840	56	197.2	8.23 (2.50–27.13)

* Multivariate model, as described in the legend of Figure 1, was also adjusted for changes in lifestyle variables. HR = hazard ratio.

DISCUSSION

In our follow-up study of more than 13 000 young men, we estimate the association of baseline values and changes in triglyceride levels with CHD. The results suggest that information on triglyceride levels at 2 time points 5 years apart are clinically relevant for assessing the risk for CHD. Among young men with triglyceride levels lower than 3.39 mmol/L (<300 mg/dL) who were not receiving lipid-lowering therapy, changes in triglyceride levels were statistically significantly associated with alterations in BMI, physical activity, and the habit of eating breakfast. These findings corroborate triglycerides as a sensitive marker of lifestyle changes. However, a substantial proportion of the CHD risk remained attributable to changes in triglyceride levels during the subsequent 5.5 years of follow-up, independent of the associated alterations in BMI and lifestyle habits, suggesting an independent cumulative effect. Collectively, these findings highlight the predictive value of follow-up triglyceride measurements for CHD risk assessment in apparently healthy young men and may assist in estimating the potential value of lifestyle interventions for the primary prevention of CHD.

The strengths of the MELANY study include its prospective design; detailed, uniform, and systematic follow-up assisted by a centralized, computerized database; the use of a well-defined "hard" clinical end point (angiography-proven CHD); the follow-up measurements of blood and lifestyle variables by using a standard questionnaire in a single center and laboratory; and the direct measurement of BMI (rather than self-reported weight and height).

Several limitations of the study warrant consideration. First, assessment of triglyceride levels at each time point was based on single measurements. Given the high level of intraindividual variability in triglyceride measurements (16, 17), the observed associations between triglycerides and CHD might underestimate the true strength of the association. Yet, the relatively large cohort may alleviate this limitation, and the association between changes in triglyceride levels and clinical variables (Figure 1) suggest that assigning persons to different triglyceride groups could not be a mere consequence of measurement error. Second, the MELANY cohort of career military personnel is a selected group of healthy young men. However, the characteristics of the population are remarkably similar to those of young men from various industrialized countries (18-22), and the relatively homogeneous environment to which participants in our study were exposed might reduce the effects of unknown confounders. Third, although changes in lifestyle variables were associated with changes in triglyceride levels, we still cannot claim a causal link between change in triglyceride level and CHD risk given the observational nature of the study. Finally, in assessing the contribution of changes in triglyceride levels to the CHD risk, we could control only for associated changes in those few lifestyle

variables that were available, which may limit the validity of concluding that change in triglyceride levels affects CHD risk independent of lifestyle factors.

Our results suggest a stronger association between triglycerides and CHD among men in their 40s compared with older-age cohorts. A recent meta-analysis of studies (mean participant age at entry, 56.6 years; mean follow-up, 12.1 years) yielded an adjusted odds ratio of 1.72 for the extreme tertiles of triglyceride levels (1). A meta-analysis of prospective studies conducted in the Asia-Pacific region (6) indicated that participants grouped in the highest fifth of triglyceride levels had a 1.8-fold greater risk for CHD. In the Framingham study (3), a 2-fold increase in CHD risk was reported in patients with triglyceride levels ranging from 2.8 to 4.5 mmol/L (250 to 400 mg/dL), compared with patients with triglyceride levels ranging from 0.6 to 1.1 mmol/L (50 to 100 mg/dL). In the Copenhagen Male Study (5), middle-aged (age >54 years) white men with fasting triglyceride levels of 1.6 mmol/L or greater (\geq 142 mg/dL) had a statistically significant adjusted risk ratio for CHD of 2.2. In the Lipid Research Clinics Follow-up Study (23), triglyceride levels showed no independent association with coronary death, but a stronger association has been noted in a subgroup of younger participants. The apparently stronger association reported in our study (adjusted hazard ratio, 4.05 when comparing men with triglyceride levels >1.9 mmol/L [>164 mg/dL] with those with triglyceride levels <0.7 mmol/L [<66 mg/dL] [Table 2]) could be explained by the younger age of the cohort, which means that the reference group was a particularly low-risk group, and by the proactive approach (angiography-proven CHD) used to screen for and diagnose CHD.

A unique element of our study is the follow-up triglyceride measurements 5 years after enrollment. This may be used as follows. If a previous triglyceride measurement is available, current values can be looked at as the time-2 determination. In this case, current high triglyceride levels are associated with a 6.8- to 9.1-fold increase in CHD risk, compared with the stable low levels. However, if current levels are intermediate or low, the past triglyceride level modifies the risk associated with current measurements: A current low triglyceride level may still indicate a 4.9-fold elevated risk for CHD if past levels were high. We used the first available triglyceride measurement as the time-1 determination. Future changes in triglyceride levels modify CHD risk at any level of the time-1 value. For example, a presently low triglyceride level would still be associated with a 3.8- or 6.8-fold increased risk for CHD if a future triglyceride level would place the patient in the low/intermediate or low/high subgroup, respectively. Furthermore, a present triglyceride value within the high tertile can be associated with a 8.23-, 6.84-, or 4.90-fold risk for CHD if a second measurement 5 years later would place an individual in the high/high, high/intermediate, or high/low subgroups, respectively, compared with those with stable low-tertile levels. Finally, a 5.4-fold increase in CHD risk

ARTICLE | Changes in Triglyceride Levels and Coronary Risk in Young Men

was observed already in men with median triglyceride levels of 1.2 mmol/L (104 mg/dL) and 1.3 mmol/L (118 mg/dL) at time 1 and time 2, respectively, compared with men with low triglyceride levels at both time points. This result highlights the association between moderately high triglyceride levels within the normal range and the development of CHD, further suggesting a cumulative effect of exposure to high triglyceride levels.

Atherosclerosis begins in childhood and progresses gradually throughout adulthood. Data from the Pathobiological Determinants of Atherosclerosis in Youth study (24) suggest that increased fatty streaks occur in the right coronary arteries of approximately 30% of apparently healthy individuals 30 to 34 years of age who died of noncardiovascular causes. Increased triglyceride levels are directly associated with atherogenic chylomicron and verylow-density lipoprotein remnants and more dense atherogenic LDL particles (25). The difference in CHD risk between the high/high and the high/low groups may suggest that decreasing triglyceride levels dramatically affects CHD risk within a relatively short period given the slow progression of CHD. As explained earlier, in this young age group, high triglyceride levels may identify those exhibiting accelerated atherosclerosis, resulting in clinically significant CHD by the mid-40s. A decrease in the rate of atherosclerosis per se, as well as other related processes with more short-term effects (such as the known interaction of triglycerides with the fibrinolytic-coagulation system [26]), may be involved. In previous randomized clinical intervention trials, the intervention affected several lipid components and lipoproteins classes simultaneously (27) rather than merely lowering triglyceride levels-this is why the isolated contribution of plasma triglyceride levels to the pathogenesis of CHD is still difficult to decipher.

Within 5 years, the median levels of triglycerides in the 3 tertiles have increased in our cohort from 0.68, 1.17, and 2.08 mmol/L (60, 104, and 184 mg/dL) to 0.79, 1.33, and 2.48 mmol/L (70, 118, and 220 mg/dL), respectively, suggesting a cohort effect. Analyses of data from the National Health and Nutrition Examination Survey (28) showed a similar increase in mean age-adjusted triglyceride levels in the U.S. population between 1988 to 1994 and 1999 to 2002, which may be attributed to increased prevalence of obesity in the U.S. population (29). Against this background, the subgroup that has experienced a medication-independent reduction in triglyceride levels during follow-up is a unique group for study. In this group, the decrease in triglyceride levels between time 1 and time 2 (the high/low group) was associated with reduced BMI, a diminished proportion of smoking, and an increase in physical activity and eating breakfast, all of which are related to adopting a healthier lifestyle (30). Thus, triglycerides might be a valuable biomarker of lifestyle (and/or weight) changes. However, the association of triglycerides with CHD risk remained statistically significant even after controlling for changes in BMI, smoking status, eating breakfast, and physical activity—suggesting that triglycerides also exert a lifestyle-independent effect on atherosclerosis.

One clinical implication of our study is in highlighting the value of integrating both previous and current values of triglycerides: Changes in triglyceride levels over time beyond those expected (that is, change represented herein by the cohort effect) modify the strength of the association of triglycerides with incident CHD, suggesting a cumulative effect. Furthermore, our study provides compelling evidence for the potential value of targeting triglyceride levels when trying to reduce CHD risk in young men. Targeted intervention trials are needed to prove that lowering triglyceride levels reduces CHD risk in young adult men.

From Sheba Medical Center, Tel-Hashomer, Israel; The S. Daniel Abraham Center for Health and Nutrition, Soroka Medical Center, and Ben-Gurion University of the Negev, Beer-Sheva, Israel; and Israel Defense Forces Medical Corps, Zrifin, Israel.

Note: Drs. Tirosh and Rudich contributed equally to this work.

Acknowledgment: The authors thank the Israel Diabetes Research Group for valuable support and advice, and Dr. M.J. Stamper, Harvard School of Public Health, for valuable discussions during the early stages of this work.

Grant Support: By Ben-Gurion University of the Negev, Beer-Sheva, Israel, and funds from the Israel Defense Forces National Budget.

Potential Financial Conflicts of Interest: None disclosed.

Requests for Single Reprints: Iris Shai, RD, PhD, Department of Epidemiology and Health Systems Evaluation, The S. Daniel Abraham International Center for Health and Nutrition, Ben-Gurion University of the Negev, PO Box 653, Beer-Sheva 84105, Israel; e-mail, irish@bgu .ac.il.

Current author addresses and author contributions are available at www .annals.org.

References

1. Sarwar N, Danesh J, Eiriksdottir G, Sigurdsson G, Wareham N, Bingham S, et al. Triglycerides and the risk of coronary heart disease: 10,158 incident cases among 262,525 participants in 29 Western prospective studies. Circulation. 2007;115:450-8. [PMID: 17190864]

2. Austin MA. Plasma triglyceride and coronary heart disease. Arterioscler Thromb. 1991;11:2-14. [PMID: 1987999]

3. Castelli WP. Epidemiology of triglycerides: a view from Framingham. Am J Cardiol. 1992;70:3H-9H. [PMID: 1466315]

4. Hokanson JE, Austin MA. Plasma triglyceride level is a risk factor for cardiovascular disease independent of high-density lipoprotein cholesterol level: a metaanalysis of population-based prospective studies. J Cardiovasc Risk. 1996;3: 213-9. [PMID: 8836866]

5. Jeppesen J, Hein HO, Suadicani P, Gyntelberg F. Triglyceride concentration and ischemic heart disease: an eight-year follow-up in the Copenhagen Male Study. Circulation. 1998;97:1029-36. [PMID: 9531248]

6. Patel A, Barzi F, Jamrozik K, Lam TH, Ueshima H, Whitlock G, et al. Serum triglycerides as a risk factor for cardiovascular diseases in the Asia-Pacific region. Circulation. 2004;110:2678-86. [PMID: 15492305]

7. Pischon T, Girman CJ, Sacks FM, Rifai N, Stampfer MJ, Rimm EB. Nonhigh-density lipoprotein cholesterol and apolipoprotein B in the prediction of coronary heart disease in men. Circulation. 2005;112:3375-83. [PMID: 16316964]

8. Shai I, Rimm EB, Hankinson SE, Curhan G, Manson JE, Rifai N, et al. Multivariate assessment of lipid parameters as predictors of coronary heart disease among postmenopausal women: potential implications for clinical guidelines. Circulation. 2004;110:2824-30. [PMID: 15492318]

9. Sharrett AR, Ballantyne CM, Coady SA, Heiss G, Sorlie PD, Catellier D, et al. Coronary heart disease prediction from lipoprotein cholesterol levels, triglycerides, lipoprotein(a), apolipoproteins A-I and B, and HDL density subfractions: The Atherosclerosis Risk in Communities (ARIC) Study. Circulation. 2001;104: 1108-13. [PMID: 11535564]

10. Stampfer MJ, Krauss RM, Ma J, Blanche PJ, Holl LG, Sacks FM, et al. A prospective study of triglyceride level, low-density lipoprotein particle diameter, and risk of myocardial infarction. JAMA. 1996;276:882-8. [PMID: 8782637]

11. Talmud PJ, Hawe E, Miller GJ, Humphries SE. Nonfasting apolipoprotein B and triglyceride levels as a useful predictor of coronary heart disease risk in middle-aged UK men. Arterioscler Thromb Vasc Biol. 2002;22:1918-23. [PMID: 12426225]

12. Dattilo AM, Kris-Etherton PM. Effects of weight reduction on blood lipids and lipoproteins: a meta-analysis. Am J Clin Nutr. 1992;56:320-8. [PMID: 1386186]

13. Szapary PO, Bloedon LT, Foster GD. Physical activity and its effects on lipids. Curr Cardiol Rep. 2003;5:488-92. [PMID: 14558992]

14. Tirosh A, Shai I, Tekes-Manova D, Israeli E, Pereg D, Shochat T, et al. Normal fasting plasma glucose levels and type 2 diabetes in young men. N Engl J Med. 2005;353:1454-62. [PMID: 16207847]

15. Bruce RA, Hossack KF, DeRouen TA, Hofer V. Enhanced risk assessment for primary coronary heart disease events by maximal exercise testing: 10 years' experience of Seattle Heart Watch. J Am Coll Cardiol. 1983;2:565-73. [PMID: 6875120]

16. Jacobs DR Jr, Barrett-Connor E. Retest reliability of plasma cholesterol and triglyceride. The Lipid Research Clinics Prevalence Study. Am J Epidemiol. 1982;116:878-85. [PMID: 7148814]

17. **Brenner H, Heiss G.** The intraindividual variability of fasting triglyceride—a challenge for further standardization. Eur Heart J. 1990;11:1054-8. [PMID: 2292251]

18. Diez Roux AV, Jacobs DR, Kiefe CI; Coronary Artery Risk Development in Young Adults (CARDIA) Study. Neighborhood characteristics and components of the insulin resistance syndrome in young adults: the coronary artery risk development in young adults (CARDIA) study. Diabetes Care. 2002;25:1976-82. [PMID: 12401742] 19. Juonala M, Viikari JS, Hutri-Kähönen N, Pietikäinen M, Jokinen E, Taittonen L, et al. The 21-year follow-up of the Cardiovascular Risk in Young Finns Study: risk factor levels, secular trends and east-west difference. J Intern Med. 2004;255:457-68. [PMID: 15049880]

20. Navas-Nacher EL, Colangelo L, Beam C, Greenland P. Risk factors for coronary heart disease in men 18 to 39 years of age. Ann Intern Med. 2001;134: 433-9. [PMID: 11255518]

21. Stamler J, Stamler R, Neaton JD, Wentworth D, Daviglus ML, Garside D, et al. Low risk-factor profile and long-term cardiovascular and noncardiovascular mortality and life expectancy: findings for 5 large cohorts of young adult and middle-aged men and women. JAMA. 1999;282:2012-8. [PMID: 10591383]

22. Urbina EM, Srinivasan SR, Kieltyka RL, Tang R, Bond MG, Chen W, et al. Correlates of carotid artery stiffness in young adults: The Bogalusa Heart Study. Atherosclerosis. 2004;176:157-64. [PMID: 15306189]

23. Criqui MH, Heiss G, Cohn R, Cowan LD, Suchindran CM, Bangdiwala S, et al. Plasma triglyceride level and mortality from coronary heart disease. N Engl J Med. 1993;328:1220-5. [PMID: 8464432]

24. McGill HC Jr, McMahan CA, Zieske AW, Sloop GD, Walcott JV, Troxclair DA, et al. Associations of coronary heart disease risk factors with the intermediate lesion of atherosclerosis in youth. The Pathobiological Determinants of Atherosclerosis in Youth (PDAY) Research Group. Arterioscler Thromb Vasc Biol. 2000;20:1998-2004. [PMID: 10938023]

25. Geurian K, Pinson JB, Weart CW. The triglyceride connection in atherosclerosis. Ann Pharmacother. 1992;26:1109-17. [PMID: 1421678]

26. **Breuer HW**. Hypertriglyceridemia: a review of clinical relevance and treatment options: focus on cerivastatin. Curr Med Res Opin. 2001;17:60-73. [PMID: 11464448]

27. Rubins HB, Robins SJ, Collins D, Fye CL, Anderson JW, Elam MB, et al. Gemfibrozil for the secondary prevention of coronary heart disease in men with low levels of high-density lipoprotein cholesterol. Veterans Affairs High-Density Lipoprotein Cholesterol Intervention Trial Study Group. N Engl J Med. 1999; 341:410-8. [PMID: 10438259]

28. Carroll MD, Lacher DA, Sorlie PD, Cleeman JI, Gordon DJ, Wolz M, et al. Trends in serum lipids and lipoproteins of adults, 1960-2002. JAMA. 2005; 294:1773-81. [PMID: 16219880]

29. Elfhag K, Rössner S. Who succeeds in maintaining weight loss? A conceptual review of factors associated with weight loss maintenance and weight regain. Obes Rev. 2005;6:67-85. [PMID: 15655039]

30. Ginsberg HN. Nonpharmacologic management of low levels of high-density lipoprotein cholesterol. Am J Cardiol. 2000;86:41L-45L. [PMID: 11374855]

EASY SLIDES

Download tables and figures as PowerPoint slides at www.annals.org.

Annals of Internal Medicine

Current Author Addresses: Dr. Tirosh: Department of Internal Medicine A, Sheba Medical Center, Tel-Hashomer, Israel.

Dr. Rudich: Department of Clinical Biochemistry, The S. Daniel Abraham International Center for Health and Nutrition, Ben-Gurion University of the Negev, PO Box 653, Beer-Sheva 84105, Israel.

Ms. Shochat and Drs. Tekes-Manova, Israeli, and Kochba: Medical Corps Headquarters, Israel Defense Forces, Zrifin, Israel.

Dr. Henkin: Soroka University Medical Center, Ben-Gurion University of the Negev, PO Box 653, Beer-Sheva 84105, Israel.

Dr. Shai: Department of Epidemiology and Health Systems Evaluation, The S. Daniel Abraham International Center for Health and Nutrition, Ben-Gurion University of the Negev, PO Box 653, Beer-Sheva 84105, Israel. Author Contributions: Conception and design: A. Tirosh, A. Rudich, D. Tekes-Manova, E. Israeli, I. Kochba, I. Shai.

Analysis and interpretation of the data: A. Tirosh, A. Rudich, I. Shai.

Drafting of the article: A. Tirosh, A. Rudich, I. Shai.

Critical revision of the article for important intellectual content: A. Tirosh, A. Rudich, Y. Henkin, I. Shai.

Final approval of the article: A. Tirosh, I. Shai.

Provision of study materials or patients: A. Tirosh, D. Tekes-Manova, E. Israeli, I. Kochba, I. Shai.

Statistical expertise: A. Tirosh, T. Shochat, I. Shai.

Obtaining of funding: E. Israeli.

Administrative, technical, or logistic support: D. Tekes-Manova, E. Israeli, I. Kochba.

Collection and assembly of data: T. Shochat.